 Vc/31-May-04
Week32 starting 31 May 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 31/5 ( ROCLA problems: 

	Bench
	Name
	Situation Mon 31/5 lunchtime
	Temp

	TBA1
	MBAR3061
	ICS 6 Disconnect Magnet WP07
	300 K

	TBA2     
	 MBAL3161    
	      New Arrival Monday Aft
	300 K

	TBB1
	MBBL1104
	ICS 5 Traveller Last
	300 K

	TBB2
	MBAL2069
	PT 12 Warm Up
	Warming

	TBC1
	MBAL3139
	PT 6.3 Training 3
	1.9 K

	TBC2
	SSS21
	MM 4 Stretched Wire/Standby
	1.9 K

	TBD1
	SSS11
	Stand By
	300 K

	TBD2
	_
	EMPTY
	 

	TBE1
	MBAL3157
	Prep 0 Scan Bench
	300 K

	TBE2
	MBBR1033
	PT 12 Warm Up
	Warming

	TBF1
	MBBL2068
	Prep 5 Pumping + Cool Down
	Cooling

	TBF2
	MBBL3163
	ICS 5 Traveller Last
	 


1 ROCLA was  in limited use. On sat 29th & Sun 30th May partial usage.  Monday 31st holiday – ROCLA not available as I was told but later I did see ROCLA on Monday +   limited ICS connect/disconnect support
Bench status  and tests for now ( Monday 31/5 evening)  & next few days : 

Test Information
 

	Magnets under Test on 31-05 7:16:31 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAR3061 
	[image: image1.png]



	14-05 
	146 h.
	108 h.
	82 h.
	19 h.
	0 h.
	ICS 6 Disconnect Magnet WP07
	28-05
(19 : 54) 
	0 A. 
	300 K
	427 hours

	2
	TBA2
	MBAL3161 
	[image: image2.png]



	31-05 
	7 h.
	-
	-
	-
	-
	Prep 0 Scan Bench
	31-05
(14 : 29) 
	0 A. 
	300 K
	7 hours

	3
	TBB1
	MBBL1104 
	[image: image3.png]



	17-05 
	26 h.
	35 h.
	282 h.
	-
	-
	ICS 5 Traveller Last
	29-05
(19 : 53) 
	0 A. 
	300 K
	343 hours

	4
	TBB2
	MBAL2069 
	[image: image4.png]



	27-05 
	49 h.
	8 h.
	31 h.
	12 h.
	1 h.
	ICS 5 Traveller Last
	31-05
(13 : 30) 
	0 A. 
	300 K
	107 hours

	5
	TBC1
	MBAL3139 
	[image: image5.png]



	27-05 
	46 h.
	29 h.
	24 h.
	5 h.
	-
	PT 12 Warm Up
	31-05
(14 : 40) 
	0 A. 
	Warming
	104 hours

	6
	TBC2
	SSS21 
	[image: image6.png]



	13-05 
	14 h.
	82 h.
	345 h.
	-
	-
	MM 4 Stretched Wire
	28-05
(13 : 47) 
	550 A. 
	1.9 K
	441 hours

	7
	TBD1
	SSS11 
	[image: image7.png]



	27-05 
	107 h.
	-
	-
	-
	-
	Stand By
	29-05
(17 : 39) 
	0 A. 
	300 K
	107 hours

	8
	TBE1
	MBAL3157 
	[image: image8.png]



	29-05 
	50 h.
	-
	-
	-
	-
	Prep 2.2 IAP @ warm LF
	31-05
(18 : 51) 
	0 A. 
	300 K
	50 hours

	9
	TBE2
	MBBR1033 
	[image: image9.png]



	27-05 
	42 h.
	18 h.
	23 h.
	14 h.
	1 h.
	ICS 5 Traveller Last
	31-05
(17 : 20) 
	0 A. 
	300 K
	100 hours

	10
	TBF1
	MBBL2068 
	[image: image10.png]



	29-05 
	31 h.
	24 h.
	-
	-
	-
	CR 1 Leak Test
	30-05
(19 : 58) 
	2 A. 
	Cooling
	55 hours

	11
	TBF2
	MBBL3163 
	[image: image11.png]



	26-05 
	123 h.
	-
	-
	-
	-
	ICS 5 Traveller Last
	30-05
(19 : 49) 
	0 A. 
	300 K
	123 hours


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR3061
	12231 Amp.
(D1_U)
	12850 Amp.
([image: image12.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3161
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL1104
	11786 Amp.
(D1-L)
	12351 Amp.
(D1-L)
	12720 Amp.
(D1_L_S13)
	12315 Amp.
(D1-U-S13)
	12472 Amp.
(D1-U-S01)
	12642 Amp.
(D1-L-S13)
	12817 Amp.
(D1-U-S02)
	12850 Amp.
(No Data)
	12850 Amp.
(No Data)
	[image: image14.jpg]



	-
	-
	-

	4
	MBAL2069
	11790 Amp.
(D2, L, S)
	12357 Amp.
(D2_L)
	12164 Amp.
(D2-L-S03)
	12211 Amp.
(D2-L-S13)
	12303 Amp.
(D2-L-S13)
	12481 Amp.
(D2_S13)
	[image: image15.png]



	-
	-
	-
	-
	-
	-

	5
	MBAL3139
	9541 Amp.
(D1_L)
	12042 Amp.
(D1-L)
	12368 Amp.
(D1 L)
	[image: image16.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS21
	12003 Amp.
(QE-34)
	12105 Amp.
(QE-12)
	12458 Amp.
(No Data)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS11
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL3157
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR1033
	12526 Amp.
(D1-U)
	0 Amp.
(D2-U)
	12850 Amp.
([image: image20.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL2068
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBL3163
	10107 Amp.
(D2-L)
	12026 Amp.
(D1-U)
	12419 Amp.
(D1-U)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-


Test Coordinator this week: Nikolay  ; MTM support on call :  (16-3602).

6 Dipoles  Disconnected after tests last week    ( Sat Midnight to Sat midnight)  
	

	Magnets under Test from 23-05-2004 to 29-05-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBB2
	MBAL1071 
	[image: image23.png]



	11-05-2004 
	52 h.
	55 h.
	23 h.
	13 h.
	21 h.
	26-05-2004 
	360 hours
	
	

	2
	TBC1
	MBBL1116 
	[image: image24.png]



	17-05-2004 
	59 h.
	42 h.
	49 h.
	23 h.
	18 h.
	27-05-2004 
	240 hours
	
	

	3
	TBD1
	MBBL2076 
	[image: image25.png]



	13-05-2004 
	14 h.
	58 h.
	42 h.
	17 h.
	31 h.
	26-05-2004 
	312 hours
	
	

	4
	TBE1
	MBBL1112 
	[image: image26.png]



	12-05-2004 
	142 h.
	0 h.
	21 h.
	13 h.
	4 h.
	29-05-2004 
	408 hours
	
	

	5
	TBE2
	MBBL1113 
	[image: image27.png]



	11-05-2004 
	51 h.
	38 h.
	58 h.
	16 h.
	35 h.
	27-05-2004 
	384 hours
	
	

	6
	TBF1
	MBBL3151 
	[image: image28.png]



	17-05-2004 
	49 h.
	31 h.
	38 h.
	16 h.
	40 h.
	29-05-2004 
	288 hours
	
	

	7
	TBF2
	MBAL3160 
	[image: image29.png]



	11-05-2004 
	46 h.
	79 h.
	46 h.
	15 h.
	11 h.
	26-05-2004 
	360 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL1071
	12509 A.
(D2-L)
	12764 A.
(D1_L)
	[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL1116
	12043 A.
(D1-L)
	12141 A.
(D1_L)
	12500 A.
(D1_L)
	12643 A.
(D1-L)
	12848 A.
(D2-L)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	MBBL2076
	11440 A.
(D1_U)
	11951 A.
(D1 L)
	12495 A.
(D1-L)
	12690 A.
(D1-L)
	12812 A.
(D1-L)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-

	4
	MBBL1112
	11808 A.
(D2-U)
	12113 A.
(D1-L)
	[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL1113
	11483 A.
(D2-L)
	11848 A.
(D2-L)
	0 A.
(No Data)
	12288 A.
(D1_U)
	12619 A.
(D2_L)
	12756 A.
(D1-L)
	12821 A.
(D2-U)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-

	6
	MBBL3151
	11485 A.
(D2-U)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL3160
	10910 A.
(D1-U)
	12568 A.
(No Data)
	12834 A.
(D1-U)
	[image: image36.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

WHY was 1033 coming again on E2  ???  tested in Feb; Perf OK ??!!  

· D1 & D2  preparation mechanically for SSS 
· SHADOW ACTIVITIES DUE TO ROCLA Problems
· Install Rotary Switch in E Cluster ( 5 days work asked) was  DONE
· Repair Water leak in F cluster Power Coverter ( 3 days asked) NOT DONE
· Calib B Cluster Power Supply in cold condition  : needs to be done once 
· Calib A     “           “            “      “     “        “               “        “          “       “
· SSS work :    AB-OP CERN Staff  daily(almost)  in new Rota    
· General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 25_N
	TBA1
	Diode test (mini NRJ at 3 kA) on 3061: Alert on QH info + Iearth = 10 mA.


	MTM
	AQA
	

	Wednesday _26_N
	TBB1
	Quench happened during the SPA on 1104:

File created with extension mg01
	MTM
	
	

	Wednesday _26_N
	TBA1
	SSL at 4.5 K. After quench TEMA blocked. On END in the window
	AB-CO
	TEMA, POWER
	TEMA relaunched.

	Thursday _27_N
	TBE2
	IAP at warm.

HF get armed and generated trigger, LF OK


	AB-CO
	HF
	Config file downloaded. MXI MOVE error , MXI COMMUNICATION error in the slots.

	Thursday _27_N
	TBE2
	IAP at warm : test of QH

 The file name in the graph did not correspond to the file name.
	AB-CO
	IAP
	Richard informed Boris. Bug corrected

	Saturday_29_A
	TBB2
	Cryo problems: During HV test , cryo HV test is lost 2 times and magnet is warming up.
	ACR
	
	

	Saturday_29_N
	TBE2
	IAP @cold on 1033. Not possible to launch the application.
	MTM
	IAP
	Maryline advised to quit all applications, stop the driver server and switch off /on the PLCSE.

	Sunday_30_M
	TBE2
	Cryo problem: Magnet blocked at 4.5 K.

Problem in the process : One valve is opening and closing always.
	ACR
	
	

	Sunday_30_M
	TBB2
	Quench antenna signals were not as expected.

Possible inversion between apertures?
	MTM
	HF
	Maryline came and solved the problem. The HF config files were changed because of the special shafts.

	Sunday_30_A
	TBE2
	Ramp to 9 T : Slow Power Abort at 12500 A. No fault detected on PLCSE.
	MTM
	
	








